Author's response to reviews

Title: Changes in medicine course curricula in Brazil encouraged by the Program for the Promotion of Changes in Medical School Curricula (PROMED)

Authors:

Patricia Alves De Souza (p.as@ig.com.br)
Angelica Maria Bicudo Zeferino (ambz@fcm.unicamp.br)
Marco Aurelio Da Ros (ros@ccs.ufsc.br)

Version: 7 Date: 18 November 2008

Author's response to reviews: see over
The BioMed Central Editorial Production Team

- Was added to the title page the title, author, affiliation, e-mail addresses and correspondence.
- As requested in the text references were cited in consecutive numbers in brackets.
- The manuscript was reviewed as the previous request, and that is divided into sections.

Regards,

Patrícia Alves de Souza.